HONG KONG UNIVERSITY OF SCIENCE & TECHNOLOGY

COMP 102 : Pr ogramming & Computer Fundamentals I

Summer 1996

Laboratory Assignment 4

22 & 24 July

Problem description :

Consider the following definitions: 1. Any integer is an operand. 2. Any operand is an expression. 3. If expression1 and expression2 are expressions, then so are

expression1 + expression2 expression1 - expression2

Writing a program to input a stream of characters ended with a < newline> character . If the input stream is an expression, evaluate the expression and ouptut the result; otherwise, output an error message. Note that spaces are allowed before and after each subexpression and each operator (+ or -).

Sample input/output:

Input expression: 345<newline> Ouput result : 345

Input expression: 3+4<newline> Ouput result : 7

Input expression: 3 + 4 - 8 <newline> Ouput result : -1

Input expression: -7 - -8 <newline> Ouput result : 1

Input expression: 78 90 <newline> Ouput result : Error

Note: You may assume that overflow and underflow will never occur in the inputs or in evaluating the expression.